Author’s response to reviews

Title: PREVENTION PRACTICES INFLUENCING FREQUENCY OF OCCURRENCE OF VASO-OCCCLUSIVE CRISIS AMONG SICKLE CELL PATIENTS IN ABEOKUTA SOUTH LOCAL GOVERNMENT AREA OF OGUN STATE, NIGERIA.

Authors:

Olorunfemi Amoran (drfamoran@yahoo.com)

AHMED JIMOH (drjimoh@yahoo.com)

Omotola Ojo (dromotolaojo@yahoo.com)

Version: 1 Date: 27 Feb 2016

Author’s response to reviews:

Dear Editor,

Pls find below then response to Editor’s comments

Response to the Editorial comments:

-1. We note that this study includes participants under the age of 16. Please detail the consent procedures for these participants. Please make it clear when parents were interviewed and when patients were interviewed. If any participants were aged under the age of 16, they must have informed parental consent. For patients aged over 16, they must have consented themselves - separate from any parental consent.

-------------------------------------------------------------

CORRECTED- ethics and consent addressed

--------

ACKNOWLEDGEMENT, AUTHORS CONTRIBUTION AND COMPETING INTREST INCLUDED

Thank you

Dr Amoran